Author's response to reviews

Title: Targeting cellular metabolism by 3-Bromopyruvate improves Tamoxifen Cytotoxicity in Breast Cancer Cell Lines

Authors:

Yasmin M Attia 1 (yasmin.attia@nci.cu.edu.eg)
Hanan S EL-Abhar 2 (hanan.elabhar@pharma.cu.edu.eg)
Mahmoud M Al Marzabani 1 (mahmoud.almarzabani@nci.cu.edu.eg)
Samia A Shouman 1 (samia.shouman@nci.cu.edu.eg)

Version: 2 Date: 22 June 2015

Author's response to reviews: see over
Dear Editor-in-Chief:

Kindly find our attached manuscript, entitled “Targeting cellular metabolism by 3-Bromopyruvate improves Tamoxifen Cytotoxicity in Breast Cancer Cell Lines” for publication in your journal *Toxicology in vitro*.

The purpose of this study was to examine the potential additive/synergistic anticancer effect of 3-BP, a glycolytic inhibitor, when combined with TAM that can be advantageous for the treatment of breast cancer. To fulfill this aim we studied the effect and the possible mechanisms of this combination using the breast cancer cell lines, viz., MCF-7 and T47D, as well as mice-bearing solid tumor. We believe that the study would be of interest to your readers.

In accordance with your journal’s policy, we confirm that the material contained in the manuscript has not been published and is not being submitted elsewhere.

The authors qualify for authorship and have no financial or personal relationships that might lead to conflict of interest. This manuscript is being submitted online in accordance with your policies for this type of submission.

Thank you for your consideration.

Best regards, Samia Shouman

Corresponding author

samiasshumsn@yahoo.com

Cancer Biology Department,

National Cancer Institute,

Cairo University